Author's response to reviews

Title: Factors important in the choice of a medical career: a Finnish national study

Authors:

Teppo J Heikkilä (teppo.heikkila@fimnet.fi)
Harri Hyppölä (harri.hyppola@kuh.fi)
Jukka Vänskä (jukka.vanska@laakariliitto.fi)
Tiina Aine (tiina.aine@uta.fi)
Hannu Halila (hannu.halila@laakariliitto.fi)
Santero Kujala (santero.kujala@fimnet.fi)
Irma Virjo (irma.virjo@uta.fi)
Markku Sumanen (markku.sumanen@uta.fi)
Kari Mattila (kari.j.mattila@uta.fi)

Version: 6 Date: 18 September 2015

Author's response to reviews:

Dear editor of BMC Medical Education,

We have now done the asked changes concerning the information about authors and detailed consent. We have added section "Authors' information" into the manuscript. We have also given details about detailed consent in the "Methods" section.

Your sincerely,

Teppo Heikkilä